Author's response to reviews

Title: A sexual health quality improvement program triples chlamydia and gonorrhoea testing rates among young people attending Aboriginal primary health care services in Australia

Authors:

Simon Graham (sgraham@kirby.unsw.edu.au)
Rebecca J Guy (rguy@kirby.unsw.edu.au)
Handan C Wand (hwand@kirby.unsw.edu.au)
John Kaldor (jkaldor@kirby.unsw.edu.au)
Basil Donovan (bdonovan@kirby.unsw.edu.au)
Janet Knox (Janet.Knox@ncahs.nsw.gov.au)
Debbie McCowen (DMccoewen@armajun.org.au)
Patricia Bullen (PBullen@wachs.org.au)
Julie Booker (jbooker@illawarraams.com.au)
Chris O'Brien (cobrien@illawarrams.com.au)
Kristine Garrett (KGarrett@galambila.org.au)
James S Ward (james.ward@sahmri.com)

Version: 2 Date: 19 January 2015

Author's response to reviews:

Dear Editorial office,

Please find attached a revised manuscript with the following changes:

1. a title page with all the authors email addresses
2. a revised abstract
3. Line and page numbering
4. a competing interests section
5. an authors contributions section
6. an acknowledgements section

Please do not hesitate to contact me if you have any questions.

Kind regards

Simon Graham